Author's response to reviews

Title: Treatment strategies in colorectal cancer patients with initially unresectable liver-only metastases, the randomised phase 3 CAIRO5 study of the Dutch Colorectal Cancer Group (DCCG)

Authors:

Joost Huiskens (j.huiskens@amc.nl)
Thomas M van Gulik (t.m.vangulik@amc.nl)
Krijn P van Lienden (k.p.vanlienden@amc.nl)
Marc R.W. Engelbrecht (m.r.engelbrecht@amc.nl)
Gerrit A Meijer (ga.meijer@vumc.nl)
Nicole C.T. van Grieken (nct.vangrieken@vumc.nl)
Jonne Schriek (j.schriek@iknl.nl)
Astrid Keijser (a.keijser@iknl.nl)
Linda Mol (l.mol@iknl.nl)
Quintus I.Q. Molenaar (i.q.molenaar@umcutrecht.nl)
Kees Verhoef (c.verhoef@erasasmusmc.nl)
Koert P de Jong (k.p.de.jong@umcg.nl)
Kees H.C. Dejong (chc.dejong@mumc.nl)
Geert Kazemier (g.kazemier@vumc.nl)
Theo M Ruers (t.ruers@nki.nl)
Hans H.W. de Wilt (hans.dewilt@radboudumc.nl)
Harm van Tinteren (h.v.tinteren@nki.nl)
Cornelis J.A. Punt (c.punt@uva.nl)

Version: 2 Date: 14 October 2014

Author's response to reviews: see over
Dear Professor McKeage,

We have revised our manuscript according to the editorial request we have received; I hereby give you a point-by-point description of the changes we have made:

1: We have provided line numbering and page numbers.

2: We have provided e-mail addresses of all authors on the title Page 1.

3: We have provided a competing interests section on page 10 of the manuscript.

4: We have provided a Acknowledgements section on page 10 of the manuscript.

5: Our EudraCT missed the first number, we have corrected that and we have also provided our ClinicalTrials.gov registration number.

Thank you very much for considering this paper for publication in BMC cancer.

Yours sincerely,

Joost Huiskens, and Kees Punt, on behalf of all authors.